Author’s response to reviews

Title: Rectal aberrant crypt foci (ACF) as a predictor of benign and malignant neoplastic lesions in the large intestine

Authors:

Marek Kowalczyk (forel@neostrada.pl)
Marcin Orłowski (m.orlowski@przychodniachelm.pl)
Łukasz Klepacki (lukasz.klepacki@uwm.edu.pl)
Krzysztof Zinkiewicz (kzinek@yahoo.com)
Waldemar Kurpiewski (wkurpiewski@interia.pl)
Dorota Kaczerska (nurdok@o2.pl)
Wiesław Pesta (wieslaw.pesta@vp.pl)
Ewa Zieliński (ewa.zielinski.bydgoszcz@wp.pl)
Piotr Siermontowski (nurdok@tlen.pl)

Version: 6 Date: 24 Jan 2020

Author’s response to reviews:

All patients, after expressing oral consent, gave their written consent to perform the study and participate in clinical trials, which was a condition for obtaining the positive consent of the Bioethical Committee for conducting clinical trials. The ethics committee would not give its consent without obtaining written and oral consent to perform the study and participation in the study expressed by patients.